Author’s response to reviews

Title: Perceptions on healthy eating, physical activity and lifestyle advice: opportunities for adapting lifestyle interventions to individuals with low socioeconomic status

Authors:

Andrea J Bukman (sandra.bukman@wur.nl)
Dorit Teuscher (d.teuscher@maastrichtuniversity.nl)
Edith JM Feskens (edith.feskens@wur.nl)
Marleen A van Baak (m.vanbaak@maastrichtuniversity.nl)
Agnes Meershoek (a.meershoek@maastrichtuniversity.nl)
Reint Jan Renes (reintjan.renes@hu.nl)

Version: 2 Date: 27 June 2014

Author’s response to reviews: see over
Dear editorial board of BMC Public Health,

Hereby we submit a revised version of our manuscript entitled: ‘Perceptions on healthy eating, physical activity and lifestyle advice: opportunities for adapting lifestyle interventions to individuals with low socioeconomic status’.

We made the following changes:
1. We included a ‘Competing interests’ section. We declared that we have no competing interests.
2. We included the contact details of the submitting author in the title page.

We would appreciate it if you would take our manuscript into consideration for the peer review process. We look forward to hearing from you.

Yours sincerely,

On behalf of all authors

A.J. Bukman, MSc.